Author's response to reviews

Title: Mycotic Aneurysm of the Posterior Tibial Artery - A rare complication of bacterial endocarditis: a case report

Authors:
- Sandeep Patel (sandeep99@doctors.net.uk)
- Nigel D'Souza (nige@doctors.net.uk)
- Shashank Gurjar (sylvurian@gmail.com)
- Jim Hewes (jimhewes@hotmail.com)

Version: 3 Date: 1 June 2008

Author's response to reviews: see over
To Whom it may concern,

We have had a second look at the history and histology in order to determine with greater accuracy the nature of the aneurysm. This has been addressed in the case report and an image from the histology included to support the case. The length of symptoms before diagnosis has been added but no organism was ever grown on blood cultures. Unfortunately we are unable to find any statistics as to the incidence of mycotic aneurysms. Grammatical changes have been made.

Mr. Sandeep Patel
MBBS MRCS